

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Tserenkhuu, Tselmuun; Kosempel, Stephen

Working Paper Life expectancy and business cycles in a small open economy

ISER Discussion Paper, No. 1263

Provided in Cooperation with: The Institute of Social and Economic Research (ISER), Osaka University

Suggested Citation: Tserenkhuu, Tselmuun; Kosempel, Stephen (2024) : Life expectancy and business cycles in a small open economy, ISER Discussion Paper, No. 1263, Osaka University, Institute of Social and Economic Research (ISER), Osaka

This Version is available at: https://hdl.handle.net/10419/306794

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Discussion Paper No. 1263

ISSN (Print) 0473-453X ISSN (Online) 2435-0982

LIFE EXPECTANCY AND BUSINESS CYCLES IN A SMALL OPEN ECONOMY

Tselmuun Tserenkhuu Stephen Kosempel

November 2024

The Institute of Social and Economic Research Osaka University 6-1 Mihogaoka, Ibaraki, Osaka 567-0047, Japan

Life expectancy and business cycles in a small open economy^{*}

Tselmuun Tserenkhuu^{ac}, Stephen Kosempel^b

^a Institute of Social and Economic Research, Osaka University, 6-1 Mihogaoka, Ibaraki, Osaka 567-0047, Japan ^bDepartment of Economics and Finance, University of Guelph, Guelph, ON N1G 2W1, Canada

Abstract

This paper examines the effects of increased life expectancy on the short-run macroeconomic stability of a typical small open economy. We develop a real-business-cycle (RBC) model of a small open economy that is consistent with the main empirical facts of economic fluctuations in open economies. Different from the previous satisfactory open-economy extensions of the baseline RBC model, given its finite lifetimes feature, our framework also helps rationalize some of the key results from recent empirical literature on the relationship between longevity and business cycles. In our model, changes in life expectancy change the planning horizon of individuals and thus affect their intertemporal choices. Consequently, the cyclical volatilities of aggregate variables are also affected. As a numerical exercise, we quantify how increased life expectancy has impacted Canadian business cycle fluctuations over the past forty years. The results indicate that the fluctuations in physical capital, human capital, and consumption all decrease as life expectancy increases. On the other hand, the fluctuation of hours worked, output, and trade balance ratio are found to increase with life expectancy.

Keywords: Business cycles, Life expectancy, Finite lifetimes, Small open economy *JEL Classification*: E32, F41, J11

 $[^]c \mathrm{Corresponding}$ author: Tselmuun Tserenkhuu, E-mail: tt
serenk@iser.osaka-u.ac.jp

Stephen Kosempel, E-mail: kosempel@uoguelph.ca

^{*}The first author gratefully acknowledges the financial support from Management Expenses Grant and Joint Usage/Research Grant by ISER, Osaka University. The authors would like to thank participants at the 56th Annual Conference of the Canadian Economics Association, the 48th Eurasia Business and Economics Society Conference, the 17th CIREQ PhD Students' Conference, and seminar participants at the University of Guelph, Saint Mary's University, and Osaka School of International Public Policy (OSIPP), for their valuable comments and discussions.

1. Introduction

Over the last two centuries life expectancy at birth has doubled in most parts of the world due to advances in medicine, public health, and living standards. While there exists a large theoreticalquantitative literature that focuses on the relationship between life expectancy and long-run economic outcomes, the theoretical link between increased longevity and the short-run macroeconomic performance is ambiguous. In part, this is because the mechanisms behind the relationship are not well understood and explored. The existing theoretical literature includes only few studies, based on overlapping generations models with rich labor market features, that focus on the effect of longevity on the business cycle strictly through changes in the labor market dynamics. However, as pointed out in Dantas Guimarães and Ferreira Tiryaki (2020), intuitively, increased longevity can potentially affect short-run macroeconomic stability through three broad channels: by changing individuals' behavior, by changing the effectiveness of fiscal and monetary policies, and by affecting labor market dynamics. Their empirical results show that the volatility of investment and, to a lesser degree, of consumption decreases as longevity rises. The authors suggest that this outcome is likely to be the result of behavioral changes—including increased aversion to risk, shifts in consumption habits, and changes in savings and financial decisions—observed as individuals grow older. On the other hand, they find that output volatility grows as longevity increases likely due to changes in labor market dynamics and greater net exports' volatility. The latter factor implies that international trade volatility is an additional channel through which longevity appears to influence output volatility.

To help rationalize some of the abovementioned empirical results, the aim of the present paper is to set up a version of the standard small open economy RBC model that formally analyzes how increased longevity affects business cycle fluctuations of an open economy through changes in the individual's intertemporal investment and labor-leisure decisions. Uncovering the potential impacts of life expectancy on economic fluctuations will contribute to our understanding of the determinants of macroeconomic volatility, which is essential for economic policymakers to design stabilization policies and for businesses and households to make informed decisions. In the context of the RBC framework, changes in life expectancy change the planning horizon of individuals and thus affect their intertemporal optimal choices. Consequently, if individuals care more about the future because they live longer, then intuitively, one can expect that the volatility of investment in different types of assets—including physical capital, human capital, and a net foreign financial asset—in the economy will be affected. In addition, when agents expect to live longer, their intertemporal substitution of hours worked and leisure may increase, thereby affecting hours worked volatility.

The model developed in the current paper is an otherwise standard RBC model of a small open economy extended with human capital accumulation and international borrowing constraints à la Barro et al. (1995) and with finite lifetimes à la Blanchard (1985), and builds primarily on the open economy neoclassical growth framework of Tserenkhuu and Kosempel (2023). In addition to allowing for mechanisms for life expectancy to affect economic outcomes, the foregoing features also help deal with the counterfactual predictions inherent in the small open economy macro models, namely, an exaggerated variability of capital stock and investment, and non-stationary equilibrium for consumption and assets. Specifically, the former two features are for slowing down convergence and dampening business cycle movements, while the latter one is for inducing stationarity of the equilibrium dynamics. To study the growth implications of changes in life expectancy from the perspective of an open economy, Tserenkhuu and Kosempel (2023) developed a deterministic continuous-time version of the model we set up in the current paper. However, since their focus was on long run outcomes, unlike the current paper, they abstracted from any shocks affecting the domestic economy in the short run and endogenous labor-leisure decisions by households. Their quantitative results revealed that life expectancy has a substantial impact on the steady state values of the key macroeconomic variables. Given the foregoing results, using the model developed in this paper, we set out to explore the business cycle implications of life expectancy in a small open economy setting. As a numerical exercise, we quantify how changes in life expectancy have impacted Canadian—an economy that has been used frequently in the macroeconomics literature as an example of a small open economy—business cycle fluctuations over the past forty years. To preview our quantitative results, we find that volatility of physical capital, human capital, and consumption all decrease as life expectancy increases. On the other hand, the volatility of hours worked, output, and trade balance ratio are found to increase as life expectancy increases. Followed from the model's feature of diminishing returns to human capital, our results also reveal that changes in life expectancy will have a greater effect—in terms of magnitude—on the volatility of the above mentioned aggregate variables the lower is the initial level of life expectancy.

It is important to study the relationship between life expectancy and macroeconomic volatility at business cycle frequencies in an open economy setting given the substantial borrowing and lending and the trade of goods and services that exist across borders of countries around the world. The widely used closed economy assumption is difficult to justify, especially if the domestic economy's behaviour is likely to be affected by its openness to the world. In fact, there exists a large empirical literature that demonstrates stronger correlations between business cycle fluctuations in real GDP for countries that trade more with one another as in Frankel and Rose (1998), Clark and van Wincoop (2001), Baxter and Kouparitsas (2005), Burstein et al. (2008), and di Giovanni and Levchenko (2010), among others. Moreover, Giovanni and Levchenko's (2009) panel data analysis of 61 countries reveals a positive and economically significant relationship between trade openness and aggregate volatility. Thus, one can expect that the link between life expectancy and volatility can also be affected by the openness of the economy. As already mentioned, Dantas Guimarães and Ferreira Tiryaki's (2020) empirical findings do support this argument.

The paper proceeds as follows. The next section covers the related literature. Section 3 introduces the model. Section 4 analyzes the steady state and the dynamics of the model. Section 5 calibrates the model. Section 6 evaluates the quantitative predictions of the model. Section 7 discusses the results regarding the effects of changes in life expectancy on the steady state levels and cyclical volatilities of the aggregate variables. We also present some quantitative results regarding the small open economy of Canada. Section 8 provides concluding remarks.

2. Related literature

In this section we review two strands of literature that are related to the current paper. As mentioned in the introduction, open-economy RBC models lead to several counterfactual predictions, namely, an exaggerated variability of capital stock and investment, and non-stationary equilibrium for consumption and assets. Thus, we first discuss some of the previous notable modifications to the standard model that are aimed at resolving the foregoing problematic results and explain why we chose to pursue the particular extensions described in the current paper. Second, our paper is also related to the literature—both theoretical and empirical—that aims to quantify the effect of demographics and on business cycle fluctuations.

To moderate the volatility of capital stock and investment, most previous open-economy RBC models have relied on capital-adjustment costs as in Cardia (1991), Mendoza (1991), Correia et al. (1995), Letendre (2004), and Letendre and Luo (2007), among many others. The main assumption here is that the cost of changing capital stock by a fixed amount increases with the speed of the desired adjustment. This gives agents an incentive to undertake investment changes gradually. Nevertheless, there are several reasons to prefer the human capital-borrowing constraint method for our analysis. First, in the growth literature human capital has been identified as an important channel for life expectancy to affect long-run economic outcomes (mostly demonstrated using closed economy models). We cannot, therefore, ignore the possibility that human capital also provides a channel for life expectancy to affect economic conditions in the short term. Second, adjustment cost functions are designed to slow transitional dynamics, but not to affect steady state outcomes. Therefore, an adjustment cost function could be used to replace the borrowing constraint, while maintaining a lower volatility of capital stock and investment, but we would still need human capital otherwise our analysis would overlook the possible impact that changes to steady state values have on short term fluctuations. And once human capital is added to a model with mortality, a borrowing constraint seems very reasonable. The reason for this is that human capital provides poor collateral, because if agents die their human capital is destroyed.

On the other hand, to induce stationarity of the equilibrium dynamics for the framework, the following techniques have been proposed in the existing literature: models with finite lifetimes—following Blanchard's (1985) perpetual youth framework—as in Leiderman and Razin (1989) and Cardia (1991); models with an endogenous discount factor as in Obstfeld (1990), Mendoza (1991), Uribe (1997), and Schmitt-Grohe (1998), among others; models with a debt-elastic interest rate premium as in Senhadji (2003), Mendoza and Uribe (2000), Schmitt-Grohe and Uribe (2001), Letendre (2004), and Letendre and Luo (2007), among others; a model with convex portfolio adjustment costs as in Neumeyer and Perri (2005); and a model with complete asset markets as in Schmitt-Grohe and Uribe (2003). Schmitt-Grohe and Uribe (2003) assess the extent to which

most of these stationarity-inducing methods¹ affect the equilibrium dynamics at business-cycle frequencies. According to their results, when calibrated with common parameter values, all models predict virtually identical unconditional second moments and impulse response functions for key macroeconomic variables. The authors conclude that if the reason for modifying the standard nonstationary small open economy model with any of the features mentioned above is simply technical, that is, solely aimed at introducing stationarity, then in choosing a particular modification of the model the researcher should be guided by computational convenience. In this respect, having had equilibrium conditions that contain an additional state variable—human capital—plus finite horizons, our framework might be in disadvantage vis-à-vis the other models. Nonetheless, we adopt the finite horizons feature since it also allows us to study the effects of changes in life expectancy captured by exogenous changes in the horizon index of our model—on the cyclical volatilities of the aggregate variables for a small open economy.

The two papers that are perhaps the most closely related to ours are Cardia (1991) and Mendoza (1991). Similar to our modelling approach, Cardia (1991) develops a small open economy RBC model with finite lifetimes à la Blanchard (1985) to explore the effects of exogenous shocks to technology, nominal monetary growth, and fiscal spending—while abstracting from shocks affecting the world real interest rate—on the dynamics of a small open economy. The key assumption in the finite horizons framework is that agents face, in each period of their life, a constant probability of death. This drives a wedge between the world real interest rate and the subjective discount rate that becomes a function of financial wealth. Accordingly, adjustments in the subjective discount rate ensure the existence of a stationary equilibrium in which the two rates are equalized. On the other hand, different from ours, to slow down the movements in capital stock and investment, as in Mendoza (1991), her model has only one type of capital, and relies on capital adjustment costs. Given the focus of the paper, her model also incorporates money via a cash-in-advance constraint as in Cooley and Hansen (1989). In addition, different from us and Mendoza (1991), her model is calibrated to the U.S. economy rather than the Canadian economy. With highly persistent productivity shocks, her model is found to replicate the observed high national saving-investment correlation and the countercyclical behavior of the current account remarkably well. Her results

¹The exception is the finite lifetimes feature. The authors do not consider this case.

suggest that most of the changes in output, saving, investment, and in the current account are due to technological shocks rather than monetary and fiscal changes.

Mendoza (1991) also extends the standard RBC model to the case of a small open economy. Different from Cardia (1991), whose objective was to study the impact of government policy shocks in a small open economy; Mendoza, like us, studies shocks to technology and the world real interest rate. The author shows that, due to the separation of saving and investment that characterizes a small open economy, the basic RBC model without capital adjustment costs performs poorly in terms of mimicking the observed business-cycle statistics of investment. In the open economy framework, when the productivity shock hits the economy, the capital stock is rapidly and freely adjusted to maintain equality of the expected marginal returns of domestic capital and foreign assets. To moderate the variability of capital accumulation and thus investment, rather than relying on imperfections in international capital mobility as in the current paper, Mendoza (1991) embeds capital-adjustment costs into the model and assumes perfect capital mobility. In addition, in the spirit of Uzawa (1968) and Obstfeld (1981), an endogenous rate of time preference—captured in the proposed stationary cardinal utility function—that increases with past consumption levels is introduced into his model to ensure a well-defined stationary equilibrium for the holdings of foreign assets. Given this preference formulation, as mentioned in the paper, the dynamics of the model works as follows: as long as the time-preference rate is smaller (greater) than the interest rate, households choose to accumulate (deplete) foreign assets in order to fund an increasing (decreasing) consumption. Once the rate of time preference adjusts to the world's real interest rate, the accumulation of external assets reaches its steady state level. Arguably, there are at least two attractions in our modelling approach compared to Mendoza (1991). First, in our framework, the effective discount rate varies in response to levels of consumption and assets and is truly endogenous in the sense that this feature materializes out of the finite horizon structure of the model, rather than simply being imposed on the utility function. Second, in our model, the horizon index can be chosen arbitrarily anywhere between zero and infinity and this allows us to study the changes in life expectancy on the behavior of the economy. In Mendoza's (1991) analysis, in contrast, the individuals are assumed to be infinitely lived.

The literature that focuses on the relationship between demographics and the business cycle is

relatively thin. By developing a closed economy overlapping generations labor search model with matching frictions and productivity shocks, Lugauer (2012a) shows how the age distribution in the population affects aggregate fluctuations and demonstrates that this effect is quantitatively large. On the empirical side, using panel data methods, Jaimovich and Siu (2009) find that changes in the age composition of the workforce account for a significant fraction of the variation in business cycle volatility observed in G7 economies. Likewise, based on standard panel data methods with lagged birth rates as an instrument, Lugauer (2012b) finds a strong statistical relationship between the relative supply of young workers and GDP volatility across the United States.

The abovementioned studies consider the effects of longevity on the business cycle strictly through changes in the labor market dynamics. However, in their recent work, Dantas Guimarães and Ferreira Tiryaki (2020) provide empirical evidence that increased longevity may also affect the volatility of macroeconomic aggregates through changes in individuals' behavior as well as through international trade volatility. Our work contributes to the ongoing research by means of a theoretical framework which explains how increased life expectancy affects the aggregate volatility of a small open economy through changes in individuals' intertemporal investment and labor-leisure decisions.

3. The Model

3.1. Households

Following Blanchard (1985), we introduce finite lifetimes into the model by assuming that agents face a constant probability of death, λ , in each period. Consequently, the survival probability from one period to the next is $1 - \lambda$. Without loss of generality, the size at birth of a cohort born at any date s is normalized to λ . Thus, the size at date t of a cohort born at date s is $\lambda(1 - \lambda)^{t-s}$, and the population size at any date t, L_t , is $\sum_{s=0}^{\infty} \lambda(1 - \lambda)^s = \frac{\lambda}{1 - (1 - \lambda)} = 1$. Consequently, the model will be used to study economywide per capita average values. The life expectancy of a member of cohort born at date s is $\sum_{t-s=1}^{\infty} (t-s)\lambda(1-\lambda)^{t-s-1} = \frac{1}{\lambda}$.

In period t, a representative agent of cohort s maximizes his expected lifetime utility of the

form:

$$E_t \left\{ \sum_{v=t}^{\infty} (\beta(1-\lambda))^{v-t} \left[\pi ln(c_{s,v}) + (1-\pi) ln(1-n_{s,v}) \right] \right\}$$
(1)

where $\beta \in (0, 1)$ is the pure rate of time preference, $\beta(1 - \lambda)$ captures the effective rate of time preference, π is the weight on consumption in utility, and $c_{s,v}$ and $n_{s,v}$ are, respectively, time-vconsumption and labor by an agent born at time s. Agents are born with no wealth and have no bequest motives. As in Yaari (1965) and Blanchard (1985), uncertainty about death and the absence of a bequest motive create the need for a life insurance market whereby agents sell contingent claims on their assets to insurance companies. Assets, except for human capital which is destroyed when the agent dies, are collected in each period from λ agents who died and transferred to the surviving agents who are of the size $1 - \lambda$. With a perfectly competitive insurance market, each surviving agent receives a premium payment of $\frac{\lambda}{1-\lambda}$ and the gross rate of return on the insurance contract is given by $1 + \frac{\lambda}{1-\lambda} = \frac{1}{1-\lambda}$. Given this gross return and assuming that agents are endowed with one unit of time that can be allocated between labor and leisure, the budget constraint in period t for an agent born at date s is given as follows:

$$c_{s,t} + k_{s,t+1} + f_{s,t+1} + h_{s,t+1} = \left(\frac{1+r_t}{1-\lambda}\right)(k_{s,t} + f_{s,t}) + (1+q_t - \delta_h)h_{s,t} + w_t n_{s,t}$$
(2)

where $k_{s,t}$, $h_{s,t}$, and $f_{s,t}$ denote, respectively, the stock of physical capital, the stock of human capital, and net foreign assets at period t; and δ_h , r_t , q_t , and w_t denote, respectively, the depreciation rate for an agent's human capital, and the real interest rate, the real rental rate of human capital, and the wage rate in period t.

As in Barro et al. (1995), an international borrowing constraint is introduced into the model by assuming that only physical capital is an acceptable collateral on foreign loans, while human capital and raw labor are not². Consequently, the foreign debt can be positive but cannot exceed

 $^{^{2}}$ The authors motivate the borrowing constraint as follows. "Physical capital is more easily repossessed and more readily monitored than human capital and is therefore more easily financed with debt. Moreover, physical capital is also more amenable to direct foreign investment: one can own a factory but not someone else's stream of labor income." The finite horizons feature of our model provides additional motivation. In the event of death, an agent's human capital is destroyed, whereas for physical capital both debt and equity can be transferred to surviving members of the economy.

the amount of physical capital:

$$-f_{s,t} \le k_{s,t}.\tag{3}$$

Note that if the borrowing constraint (3) is slack, then the economy has sufficient collateral to borrow and will adjust both types of capital rapidly. This situation would lead to counterfactually high transitional growth and investment volatility. Therefore, to avoid the exaggerated variability of capital and having instantaneous convergence to the balanced growth path, we restrict our attention to the case where the borrowing constraint is binding:

$$-f_{s,t} = k_{s,t} \quad \forall t. \tag{4}$$

Given (4), the flow budget in (2) can now be written as follows:

$$c_{s,t} + h_{s,t+1} = (1 + q_t - \delta_h)h_{s,t} + w_t n_{s,t}.$$
(5)

In the spirit of Frenkel and Razin (1986), for the purpose of deriving the present value at date t of agent's lifetime budget constraint, we define the following present-value factor, τ_t , which is composed of one-period real rental rates of human capital compounded from period 1 to period t:

$$\tau_t = \{(1+q_1-\delta_h)(1+q_2-\delta_h)...(1+q_t-\delta_h)\}^{-1}.$$
(6)

Therefore, τ_t/τ_{t+1} equals $(1 + q_{t+1} - \delta_h)$, which is the gross rate of return to individual human capital stock in period t + 1. Given τ_t and the survival probability from one period to the next, $1 - \lambda$, the present-value lifetime budget constraint can be derived by consolidating the periodic budget constraints for the current period t and the future periods (i.e., periods t+1, t+2, ...):

$$E_t \sum_{v=t}^{\infty} (1-\lambda)^{v-t} \frac{\tau_v}{\tau_t} c_{s,v} = E_t \left(\sum_{v=t}^{\infty} (1-\lambda)^{v-t} \frac{\tau_v}{\tau_t} w_v n_{s,v} + (1+q_t-\delta_h) h_{s,t} \right).$$
(7)

Note that, in deriving equation (7), we have imposed the transversality condition given by:

$$\lim_{v \to \infty} (1 - \lambda)^v \tau_v h_{s,v} = 0.$$
(8)

The agent's problem is to maximize the expected lifetime utility in equation (1) subject to the lifetime budget constraint in (7), and the Lagrangian associated with the optimization problem can be written as follows:

$$\mathcal{L} = E_t \sum_{v=t}^{\infty} (\beta(1-\lambda))^{v-t} [\pi ln(c_{s,v}) + (1-\pi)ln(1-n_{s,v})] + \gamma E_t \left[\sum_{v=t}^{\infty} (1-\lambda)^{v-t} \frac{\tau_v}{\tau_t} w_v n_{s,v} + (1+q_t-\delta_h)h_{s,t} - \sum_{v=t}^{\infty} (1-\lambda)^{v-t} \frac{\tau_v}{\tau_t} c_{s,v} \right]$$
(9)

where γ is the Lagrange multiplier associated with the budget constraint. The first order conditions with respect to $c_{s,v}$ and $n_{s,v}$, respectively, imply:

$$c_{s,v} = E_t \left[\beta^{v-t} \pi \left(\gamma \frac{\tau_v}{\tau_t} \right)^{-1} \right] \quad \text{for} \quad v = t, t+1, t+2, \dots$$
 (10)

$$n_{s,v} = E_t \left[1 - \beta^{v-t} (1-\pi) \left(\gamma \frac{\tau_v}{\tau_t} w_v \right)^{-1} \right] \quad \text{for} \quad v = t, t+1, t+2, \dots$$
(11)

Using equations (7) and (10), we can express consumption at time t as a function of wealth as follows:

$$c_{s,t} = \Delta^{-1} E_t \left[\sum_{v=t}^{\infty} (1-\lambda)^{v-t} \frac{\tau_v}{\tau_t} w_v n_{s,v} + (1+q_t-\delta_h) h_{s,t} \right]$$
(12)

where $\Delta \equiv \sum_{v=t}^{\infty} ((1-\lambda)\beta)^{v-t} = \frac{1}{1-(1-\lambda)\beta}$; and Δ^{-1} and $\widetilde{w}_t \equiv \sum_{v=t}^{\infty} (1-\lambda)^{v-t} \frac{\tau_v}{\tau_t} w_v n_{s,v}$ denote, respectively, the propensity to consume out of wealth and the present value at date t of the agent's lifetime wage income. From (10) and (11), we derive:

$$n_{s,t} = 1 - \frac{(1-\pi)c_{s,t}}{\pi w_t}.$$
(13)

3.2. Firms

A representative firm in the perfectly competitive market produces output, Y_t ,³ according to the following Cobb-Douglas technology with physical capital, K_t , human capital, H_t , and labor, N_t , as factor inputs :

$$Y_t = z_t K_t^{\alpha} H_t^{\eta} (A_t N_t)^{1-\alpha-\eta} \tag{14}$$

where A_t and z_t are the deterministic level of technology and the stochastic total factor productivity shock, respectively, and they evolve exogenously as follows:

$$A_t = A_0 e^{xt} \tag{15}$$

$$lnz_t = \rho_z lnz_{t-1} + \varepsilon_{z,t}, \quad \varepsilon_{z,t} \sim i.i.d. \ N(0, \sigma_z^2).$$
(16)

We perform a balanced growth transformation and define variables in effective worker units using lowercase letters with hats $(\wedge)^4$. Consequently, equation (14) can be expressed in its intensive form as follows:

$$\hat{y}_t = z_t \hat{k}_t^\alpha \hat{h}_t^\eta N_t^{1-\alpha-\eta}.$$
(17)

The firm's profit maximization implies that factors of production are paid their marginal products and that no arbitrage opportunities exist between physical capital and foreign assets:

$$w_t = (1 - \alpha - \eta) A_t \frac{\hat{y}_t}{N_t} \tag{18}$$

$$q_t = \eta \frac{\hat{y}_t}{\hat{h}_t} \tag{19}$$

$$r_t = \alpha \frac{\hat{y}_t}{\hat{k}_t} - \delta_K \tag{20}$$

where δ_K denotes the rate of depreciation for physical capital.

To capture transmission of technology shocks across countries, we assume that the real rental rate of physical capital faced by the agents in the small open economy, $r_t + \delta_K$, consists of three

³Aggregate variables are denoted by capital letters.

⁴Note that, since the population size at any date t, L_t , is equal to 1, the aggregate and per worker/average variables are the same in the model. Thus, $\hat{k}_t \equiv \frac{K_t}{A_t L_t} = \frac{K_t}{A_t}$.

components, a constant corresponding to the average value, R, a random disturbance affecting the world's real interest rate, g_t , and the stochastic total factor productivity shock, z_t . Note that in closed economy models the technology variable and the real interest are strongly correlated, and this is because a positive (negative) shock will raise (lower) the rate of return to an investment in physical capital, and consequently the optimal reallocation of domestic savings requires a rise (fall) in the interest rate to clear the market. In comparison, a key assumption about small open economies is that the interest rate is invariant to domestic savings and investment activities. Therefore, in order for interest rates to be procyclical, as observed in the data, business cycle activity will need to be synchronized internationally. In fact, there is a large literature that demonstrates stronger correlations between business cycle fluctuations in real GDP for countries that trade more with one another as in Frankel and Rose (1998), Clark and van Wincoop (2001), Baxter and Kouparitsas (2005), Burstein et al. (2008), and di Giovanni and Levchenko (2010), among others. Thus, we assume the following functional form $F(\cdot)$ for $r_t + \delta_K$:

$$r_t + \delta_K = F(r, z_t, g_t) = R z_t^\theta g_t^{1-\theta}$$
(21)

where $0 \le \theta \le 1$ captures the degree of transmission of technology shocks across countries. Accordingly, higher θ implies stronger synchronization of business cycle movements across countries. Moreover, g_t is assumed to evolve exogenously as follows:

$$lng_t = \rho_g lng_{t-1} + \varepsilon_{g,t}, \quad \varepsilon_{g,t} \sim i.i.d. \ N(0, \sigma_g^2).$$
(22)

Combining (20) and (21), we derive:

$$\hat{k}_t = \frac{\alpha \hat{y}_t}{R z_t^\theta g_t^{1-\theta}}.$$
(23)

From (17) and (23):

$$\hat{y}_t = z_t^{\frac{1}{1-\alpha}} \left[\frac{\alpha}{R z_t^{\theta} g_t^{1-\theta}} \right]^{\frac{\alpha}{1-\alpha}} \hat{h}_t^{\frac{\eta}{1-\alpha}} N_t^{\frac{1-\alpha-\eta}{1-\alpha}}.$$
(24)

Note that $0 < \alpha + \eta < 1$ implies that $0 < \frac{\eta}{1-\alpha} < \alpha + \eta < 1$. Consequently, from (24), we can see

that our model features diminishing returns with respect to human capital.

3.3. Competitive equilibrium

We now combine the behavior of households and firms to characterize the structure of a competitive market equilibrium. For each time period t, the competitive equilibrium consists of an allocation $\{\hat{c}_{s,t}, \hat{h}_{s,t}, \hat{k}_{s,t}, \hat{f}_{s,t}, \hat{a}_{s,t}\}$ for all living households of each cohort $s \leq t$, a production plan $\{Y_t, H_t, N_t, K_t\}$ for each firm, and a set of prices $\{r_t, w_t, q_t\}$, such that:

(i) given prices, the allocations for the households solve their optimization problem given by (1)-(8);

(ii) given prices, the allocations for the firms solve their profit maximizing conditions given by (18)-(20);

(iii) all markets clear; below -F(t) stands for the aggregate foreign debt of the domestic economy:

$$N_t = \sum_{s=0}^{\infty} \lambda (1-\lambda)^s n_{s,t}$$
(25)

$$H_t = \sum_{s=0}^{\infty} \lambda (1-\lambda)^s \hat{h}_{s,t}$$
(26)

$$K(t) = -F(t) \tag{27}$$

$$C_t + H_{t+1} = (1 - \alpha)Y_t + (1 - \delta_h - \lambda)H_t + w_t N_t;$$
(28)

(iv) individual decisions are consistent with aggregate outcomes. As in Blanchard (1985), aggregate variables can be found as the summation across cohorts, weighted by their respective sizes. For example, for aggregate consumption, $C_t = \sum_{s=0}^{\infty} \lambda (1-\lambda)^s c_{s,t}$, and so on. Using the foregoing equations that connect aggregate to individual outcomes, the first order conditions to the individuals' and firms' optimization problems, and the market clearing conditions, we can derive expressions for the aggregate resource constraint, the Euler equation for the accumulation of human capital, and the efficiency condition for aggregate hours worked, and these are given, respectively by:

$$(1 - \alpha)\hat{y}_t = \hat{c}_t + e^x \hat{h}_{t+1} - (1 - \delta_h - \lambda)\hat{h}_t$$
(29)

$$\hat{c}_{t} = \frac{\Delta^{-1}\lambda}{\beta(1-\lambda)} \left(\hat{h}_{t} + e^{x} \hat{h}_{t+1} \right) + \frac{e^{x}}{\beta} E_{t} \left\{ \frac{\hat{c}_{t+1}}{1 + \eta \frac{\hat{y}_{t+1}}{\hat{h}_{t+1}} - \delta_{h}} \right\}$$
(30)

$$N_t = 1 - \frac{(1-\pi)\hat{c}_t}{\pi(1-\alpha-\eta)\frac{\hat{y}_t}{N_t}}.$$
(31)

4. Steady state and dynamics

In this section, we characterize the steady state and dynamics of the model. In order for a stable steady state to exist for the model, the variables in equations (29)-(31) must be stationary in the long run. To study the nonstochastic steady steady, we set $\hat{c}_t = \hat{c}^*$ and $\hat{h}_t = \hat{h}^*$ for all t. Then, from the Euler equation in (30), the required condition for having consumption per effective unit of labor to be constant in the long run is given by:

$$1 + \eta \frac{\hat{y}^*}{\hat{h}^*} - \delta_h = \left[\frac{\beta}{e^x} - \frac{\Delta^{-1}\lambda}{1 - \lambda} \frac{(1 + e^x)}{e^x} \frac{\hat{h}^*}{\hat{c}^*} \right]^{-1}.$$
 (32)

Notice that the term on the left hand side of equation (32) is the rate of return to human capital and it is decreasing in \hat{h} , whereas the term on the right hand side of the equation is the steady state effective discount rate of the domestic economy and, for $\lambda \neq 0$, it can be shown to be increasing in \hat{h} . Due to the model's finite lifetimes feature, adjustments in the ratio $\frac{\hat{h}}{\hat{c}}$ in the effective discount rate help to ensure that the long run value for the rate of return to human capital will be above the steady state world interest rate, r^5 . Recall that the model equations (29)-(31) above were derived under the assumption of a binding debt constraint—given by equation (4)—and this assumption is only valid if the rate of return to human capital remains above r. Therefore, when analyzing the dynamics of the model, there are two cases to consider: (i) $1 + \eta \frac{\hat{y}^*}{\hat{h}^*} - \delta_h > r$ and (ii) $1 + \eta \frac{\hat{y}^*}{\hat{h}^*} - \delta_h = r$. We note that the case (ii) implies a situation in which the borrowing constraint no longer holds with equality. If during the transition to the steady state the accumulation of human capital causes its return to fall and equal the world real interest rate before equation (32) was satisfied—so the case (ii)—then agents would stop investing in human capital and would start accumulating foreign assets or build up equity from physical capital. Our companion paper, Tserenkhuu and Kosempel

⁵Note that, from equation (21), $r \equiv R - \delta_k$.

(2023), discusses how to deal with this situation. Specifically, once the rate of return to human capital equals r, then the rate of return to human capital and human capital are replaced in the Euler equation with r and a new variable—say \hat{b} , where \hat{b} denotes the sum of all assets: $\hat{b} = \hat{k} + \hat{h} + \hat{f}$. As the economy accumulates more assets, the effective discount rate rises until it equals r, and (32) is satisfied. In this case, having the effective discount rate rise with the assets to consumption ratio, which materializes out of the finite-horizons structure, is critical to guarantee a stable outcome for consumption and assets. Note, however, that our calibration exercise in the next section produces parameter values for which the borrowing constraint in the model is permanently binding, and thus we consider only the case (i) in the rest of this paper. Readers will find a detailed analysis of the case (ii) in our companion study which focuses on the transitional dynamics and long term outcomes in a deterministic version of this model.

The steady state consumption to human capital and output to human capital ratios— $\frac{\hat{c}^*}{\hat{h}^*}$ and $\frac{\hat{y}^*}{\hat{h}^*}$, respectively—can be found by solving equations (29) and (30) when evaluated at the steady state. Given N^* and a positive unique root for $\frac{\hat{y}^*}{\hat{h}^*}$, equation (24) at the steady state implies:

$$\hat{h}^* = \left[\left(\frac{\alpha}{R}\right)^{\alpha} \left(\frac{\hat{y}^*}{\hat{h}^*}\right)^{\alpha-1} \right]^{\frac{1}{1-\alpha-\eta}} N^*.$$
(33)

Next, given $\frac{\hat{c}^*}{\hat{h}^*}$, \hat{h}^* , and N^* , we can derive the steady state levels of consumption and output per effective labor, and the domestic saving rate for human capital, s^* , as follows:

$$\hat{c}^* = \left(\frac{\hat{c}^*}{\hat{h}^*}\right)\hat{h}^* \tag{34}$$

$$\hat{y}^* = \left[\frac{\alpha}{R}\right]^{\frac{\alpha}{1-\alpha}} \hat{h}^{*\frac{\eta}{1-\alpha}} N^{*\frac{1-\alpha-\eta}{1-\alpha}}$$
(35)

$$s^* = 1 - \frac{\hat{c}^*}{(1-\alpha)\hat{y}^*}.$$
(36)

To analyze the dynamics of the model, we log-linearize the model equations (29)-(31) around its deterministic steady state and solve the resulting system for the recursive equilibrium laws of motion

following Uhlig's (2001) method of undetermined coefficients. Once the law of motions for human capital, consumption, and labor are known and given the initial values $\{\hat{h}_0, z_0, g_0\}$ and a sequence of disturbances $\{\varepsilon_{z,t}, \varepsilon_{q,t}\}_{t=0}^{\infty}$, we can then derive the dynamic time paths for all model variables.

5. Calibration

To derive the numerical solutions to the model and evaluate the quantitative predictions of the model, it is necessary to assign values for the model parameters. We calibrate the model to Canadian quarterly data and values for most parameters are available from the existing literature. The share of physical capital in output is set at $\alpha = 1/3$ and the rate of depreciation for physical capital is set at $\delta_K = 0.025$. These values are commonly used in models calibrated to small open economies as in Mendoza (1991) Schmitt-Grohe and Uribe (2003), Letendre (2004), Nason and Rogers (2006), and Letendre and Luo (2007), among others. The pure rate of time preference is set at $\beta = 0.989$ to target the average quarterly rate of return to human capital of 2% as in the data⁶. The value of $\beta = 0.989$ is also very close to 0.99—a value estimated by Prescott (1986).

In our benchmark calibration, life expectancy is set at 74 years or 296 quarters (i.e. $\lambda = 1/296$) to match life expectancy at birth in Canada at the start of our sample. Then, in Section 7, we study how using the end of period value (82 years or 328 quarters) impacts the results.

Based on Mendoza (1991) and Schmitt-Grohe and Uribe (2003), who use the value suggested by Kydland and Prescott (1982) and Prescott (1986) for the real interest rate in the U.S. economy, we set the steady state world's real interest rate at $r = R - \delta_K = 0.01$, which implies an annual real interest rate of four percent. The value of four percent is also consistent with the estimates suggested by Hansen (1985) and McGrattan and Prescott (2001).

As mentioned in Alessandrini et al. (2015), regarding the depreciation rate for human capital, there is no agreement in the literature and estimates vary in the range 0.5% - 5% per annum. As a compromise, we choose the midpoint of the range and set δ_h at its quarterly equivalence of 0.6875%. Similarly, regarding the human capital share parameter in the production function (η) , estimates used are typically in the range 1/3 - 2/3. For example, Mankiw et al. (1992) suggest

 $^{^{6}}$ Psacharopoulos and Patrinos (2018) estimate that the private return to schooling is 8% annually for high income countries like Canada.

that a production function that is consistent with their empirical work has equal factor shares of 1/3 across the three inputs: K, H, N. While their estimate for α is consistent with physical capital's share of income as measured in the National Accounts, their value of η seems low. For example, in the endogenous growth literature, η is typically set to 2/3, implying an income share of unimproved labor (N) of zero⁷. Empirical support for using a near zero share of unimproved labor is provided by Gundlach (1995), who re-examines the work of Mankiw et al., but using a more broadly defined measure of human capital. In the current paper, as a compromise, we choose the midpoint of $\eta = 1/2$.

As it is a common practice in the RBC literature, the weights on consumption and leisure, π and $1 - \pi$ are set to target the hours worked in the steady state to 1/3, a value that is routinely used in the RBC literature. The resulting values are $\pi = 0.5449$ and $1 - \pi = 0.4551$.

Given parameter values for α and η and using Canadian quarterly data for the period 1976Q1 to 2019Q4—from CANSIM by Statistics Canada—on GDP, business gross fixed investment in nonresidential structures, machinery and equipment⁸ (a proxy for investment in physical capital), total expenditures on education⁹ (a proxy for investment in human capital), and labor hours, we construct a series of Solow residuals, *SR*, as follows:

$$SR_t = lnY_t - \alpha lnK_t - \eta lnH_{t-1} - (1 - \alpha - \eta)lnN_t.$$
(37)

According to the production function in equation (14), the above Solow residuals are related to the total factor productivity shock, z_t , and the growth rate in deterministic level of technology, x, as follows:

$$SR_t = (1 - \alpha - \eta)lnA_0 + (1 - \alpha - \eta)x \cdot t + lnz_t.$$
(38)

Based on equation (38), we regress the SR_t on a time trend, t, to find estimate of x. The estimate

⁷This includes RBC models with endogenous growth as in Gomme (1993).

⁸Given the series, we can derive the series for stock of physical capital as follows. First, we assume that the economy is on a balanced growth path and thus flow and stock of physical capital are growing on average at the same rate. Then, given this assumption and given parameter value for δ_K , the series for stock of physical capital capital capital capital capital capital capital capital capital is physical capital in period t + 1.

⁹The series for the stock of human capital is derived following the same procedure used to derive the stock of physical capital. This time, the law of motion for human capital, which is derived from equation (39), is given by: $H_{t+1} = I_{h,t} + (1 - \delta_h - \lambda)H_t$, where $I_{h,t}$ is the investment in human capital in period t.

is found to be x = 0.0041, which corresponds to an annual growth rate of 1.64%. Note that the constant, A_0 , is simply a scaling factor and can be normalized to 1.

Next, given the series on GDP and physical capital, to estimate the parameter for the degree of transmission of technology shocks across countries, θ , based on equation (20), we first construct a series for natural logarithm of the real rental rate of physical capital as follows:

$$ln(r_t + \delta_K) = ln\left(\frac{\alpha \hat{y}_t}{\hat{k}_t}\right).$$
(39)

According to equation (21), the above series for the real rental rate of physical capital is related to the total factor productivity shock, z_t , and the random disturbance affecting the world's real interest rate, g_t , as follows:

$$ln(r_t + \delta_K) = lnR + \theta lnz_t + (1 - \theta)lng_t.$$
(40)

Based on equation (40), we regress the series on the left hand on a constant and the residuals from equation (38) to find an estimate of θ , which is found to be $\theta = 0.9598$. The value of 0.9598 corresponds to a high degree of transmission of technology shocks, thereby capturing the strong correlations between business cycle fluctuations in real GDP for the small open economy and its trading partners as supported by the literature. While we consider the aforementioned high degree of transmission of technology shocks in our baseline calibration, when we present our numerical results in Section 6, we will test the sensitivity by using low and mid values for θ as well.

Next, while the residuals from the regression in (38) can be regressed on their first lag to derive estimates for the value of the autocorrelation coefficient in the stochastic total factor productivity process, ρ_z , and the standard deviation of the innovation to the process, σ_z , there are at least two reasons why this approach might not be appropriate for our analysis¹⁰. First, as already highlighted in Mendoza (1991), Solow residuals obtained from Canadian data exhibit too much variability and serial autocorrelation for the model to be consistent with the stylized facts. Second,

¹⁰According to our regression results, we find $\rho_z = 0.9887$ and $\sigma_z = 0.748\%$. While the value of 0.748% is in the range 0.7% – 1% that is suggested in the literature (See Hansen 1985, for example), the value of 0.9887 is higher than 0.95, a value that is often used in the RBC literature.

in constructing the series for investment in human capital in (37), we converted our raw annual data on total expenditures on education to quarterly frequency by interpolating linearly between points and this procedure will add additional persistence to the stock of human capital and also to the Solow residuals. Therefore, to avoid these issues and to help facilitate a comparison with existing RBC studies, we choose to adopt the following standard values used in the literature¹¹: $\rho_z = 0.95$ and $\sigma_z = 0.763\%$.

Regarding the value of the autocorrelation coefficient in the process for stochastic world interest rate, ρ_g , and the standard deviation of the innovation to the process, σ_g , there is no agreement in the literature. For example, to facilitate his numerical analysis by restricting the set of free parameters to be specified, for simplicity, Mendoza (1991) sets $\rho_g = \rho_z$. For σ_g , he considers various different values—in the range $0.678\% - 3.39\%^{12}$ —to explore his model's sensitivity to international disturbances. In our baseline calibarion, we set $\rho_g = \rho_z (= 0.95)$ following Mendoza, and the value for the parameter σ_g is set at 1.567% to mimic the volatility of investment in physical capital in the data. While $\sigma_z = 0.763\%$ and $\sigma_g = 1.567\%$ are our preferred values, in Section 6.5, we will demonstrate the sensitivity of the model to the shocks by analyzing how the results change in the absence of one shock, that is, when the volatility of one shock is set to zero. Finally, since the parameter θ was already introduced to account for the potential international transmission of technology shocks, we consider mutually uncorrelated disturbances, that is, we set $\rho_{z,g} = 0$.

6. Numerical results: The model properties

6.1. Transitional dynamics in the absence of shocks

Figure 1 shows the time paths of the model variables in the absence of any shocks (i.e. $z_t = z^* = 1$ and $g_t = g^* = 1$ for all t) and when the hypothetical economy's initial level of human capital per effective labor is set at $\hat{h}_0 = 0.25 \cdot \hat{h}^*$. Setting $\hat{h}_0 < \hat{h}^*$ allows us to study the transitional dynamics of the model; however, the actual level of \hat{h}_0 is set arbitrarily. From panels A and B, we can see that output per capita increases during the transition while the growth of output per

¹¹See Prescott (1986), for example.

¹²These correspond to the quarterly equivalents of the annual values used by Mendoza (1991).

capita decreases. These results are consistent with the evidence that the growth rate declines as income increases.

As can be seen from panels C and G, in our model, both the domestic saving rate and aggregate hours worked decrease during the transition until they reach their respective steady state values. Higher values of hours worked and savings early on in the model imply that agents in the economy are taking advantage of high returns to human capital that exist during the transition by working longer and saving to accumulate more human capital for later on when returns are lower.

Panels D, E, F, and H show, respectively, that consumption, human capital, physical capital, and foreign debt per effective labor all increase gradually during the transition until they reach their respective positive steady state values. The gradual increase in human capital during the transition reflects the feature of diminishing returns to human capital in our model, which can be seen from the production function in its intensive form in equation (24). Moreover, due to the constraint of domestic saving on the accumulation of human capital and the complementarity between the two types of capital in the Cobb-Douglas production function, the physical capital stock and level of foreign debt also rise gradually towards their steady-state values. We also see that transition paths for the variables measured in effective labor units all have concave shapes. Over time growth slows and values approach positive constants. This stability is achieved due to the model's finitelifetimes feature. The observed slow transition to the steady state is an implication of the borrowing constraint, which is critical to prevent capital stocks from adjusting instantaneously to their long run values.

6.2. Total factor productivity shocks

Figure 2 shows the impulse response functions for the aggregate variables to a positive one percent TFP shock. The graphs represent the percent deviation of each variable from steady state after the shock. We can see that all the reported variables are procyclical. The results are consistent with the standard results in the RBC literature. A positive TFP shock has a direct positive impact on output via the production function. Consequently, marginal products of all factors of production increase. Given these increases, the agents are encouraged to accumulate more human capital and increase their labor supply. The procyclical transitional dynamics for physical capital stock and



Figure 1: Time paths of the model variables in the absence of shocks

the interest rate following the TFP shock are determined by equations (20) and (21). Recall that the world interest rate will respond positively to a productivity shock, and the strength of the response will depend on the parameter θ in equation (21), which restricts/permits the international transmission of productivity shocks. Having $\theta > 0$ is necessary to reproduce procyclical movements in the interest rate. We will revisit this result in Section 6.5 when we test the sensitivity our our results to low, medium, and high values for theta. Next, given that consumption is a normal good, it also increases following the positive shock. Finally, we can see that all the variables eventually around 40-140 quarters after the shock—converge back to their steady state values.

6.3. Interest rate shocks

Figure 3 shows the impulse response functions for the aggregate variables to a positive one percent shock to the exogenous component of the world interest rate, g. We can see that all the variables are affected negatively by the shock. A positive interest rate shock makes borrowing from abroad more expensive and thus lowers the amount of foreign loans for the domestic economy. Since foreign loans require physical capital as a collateral, the domestic economy's physical capital stock also decreases following the positive interest rate shock. Moreover, since physical capital is the only acceptable form of collateral, it responds directly to the shock and is the most sensitive. Other variables will respond to the shock, but only indirectly via its impact on physical capital. For example, a fall in physical capital will in turn lead to decreased marginal products of human capital and labor. As a result, the agents accumulate less human capital and decrease their labor supply. Consequently, the output level also decreases. Since consumption is a normal good, it is also affected negatively by the positive interest rate shock. Lastly, owing to the model's finite-lifetimes feature, we can see again that all the key variables eventually—around 40-140 quarters after the shock—converge back to their steady state values.

6.4. Business cycle statistics

Table 1 reports the quarterly business cycle statistics for Canadian economy and three different versions of our model economy: (i) Model with TFP shock only; (ii) Model with interest rate shock only; and (iii) Model with both types of shocks. For the model economies, the average business cycle statistics are computed from 1,000 simulations and each simulation consists of 175 periods



Figure 2: Impulse response functions for the aggregate economy, a TFP shock



Figure 3: Impulse response functions for the aggregate economy, an interest rate shock

as in the data. As it is standard, before computing the statistics, each simulated series was logged and detrended using the Hodrick-Prescott filter with the smoothing parameter of 1,600, which is an appropriate value for quarterly frequency data. We can see that the fluctuations in all the reported variables except for savings and trade balance ratio are larger for the actual economy than for the model economies. However, a care must be taken in making comparisons. As pointed out in Hansen (1985), there are likely to be measurement errors and sampling errors that may lead to the much higher fluctuations of the actual economy. The much lower fluctuation of consumption series in the model economies than the actual economy is the typical result for the RBC models that do not distinguish between consumer durables and non-durables. In addition, the much lower variability of hours worked in the model economies is also a well-known issue for the RBC literature, especially in models where labor supply is perfectly divisible. Despite these shortcomings, overall, the model economies (i) and (iii) are both consistent with the low volatility of consumption and large volatility of investment in physical capital relative to that of output. Further, they are in line with the high positive correlation of consumption, savings, hours worked, and investment in physical capital with output. The model economies (i)-(iii) are also able to capture the counter-cyclicality of trade balance ratio satisfactorily.

Comparisons of the model economies (i)-(iii) also reveal that fluctuations in our framework are mainly driven by TFP shocks rather than interest rate shocks. In our model, due to the borrowing constraint, the effects of interest rate disturbances on the domestic economy operate only indirectly through their effect on physical capital—which can be seen from equation (23). The relative neutrality of interest rate shocks is supported by the work of Mendoza (1991). His model also predicts that interest rate shocks have little effect on the economy. Mendoza points out that in models calibrated to the small open economy of Canada where the average interest rate is low and foreign interest payments are a small fraction of GDP, the effects generated by these shocks tend to be small. However, the author further points out that, in economies with a higher debt-service ratio, such as the heavily indebted developing countries, interest rate shocks are likely to play a more significant role.

In the model, a rise in the world interest rate has a negative effect on output, and this can be seen from the negative correlation coefficient reported for capital productivity in Model (ii)—when only interest rate shocks are present, and from the impulse response functions reported earlier. However, in the Canadian data capital productivity is strongly procyclical. Since small open economies are assumed to be price takers in international financial markets, in order to match this feature of the data, it is necessary that in the model we account for the international synchronization of business cycle activity, or more specifically, the positive correlation observed between productivity and the rate of return to physical capital. In this way, the dominate shock—productivity (z)—in the model moves both output and the world real interest rate in the same direction. Therefore, these results provide additional support for setting the degree of international synchronization (θ) to be positive. In the next subsection, we provide additional support for setting θ to be a high value.

Finally, as reported in the last row of the table, the model economies capture well the observed positive correlation between domestic savings and investment in physical capital. Having two types of capital helps to achieve this result. For example, following a positive technology shock, the rate of return for both types of capital increases. As a result, agents borrow from abroad to finance physical capital investment, and domestic savings rises to support human capital investment.

6.5. Sensitivity analysis

In this subsection, we test the sensitivity of some of our results using different values for parameters capturing the degree of transmission of technology shocks across countries, θ , the standard deviation of the innovation to the stochastic total factor productivity process, σ_z , and the standard deviation of the innovation to the process for stochastic world interest rate, σ_g . Based on equation (40), we estimated θ to be 0.9598, corresponding to a high degree of international transmission of technology shocks as supported by the literature. In Table 2, we report how the volatility of the two main macro aggregates—output and investment—changes when we use low ($\theta = 0$) and mid ($\theta = 0.5$) values for θ while keeping σ_z and σ_g at their benchmark values of 0.763% and 1.567%, respectively. The table illustrates the role of a higher degree of international transmission of technology shocks in lowering investment volatility in the model. We can see that, while the volatility of output (and other key macro aggregates not reported in the table) increases moderately with low and mid values for θ , the volatility of investment jumps to a very high level of 47.51% with $\theta = 0.5$, and further to 93.01% with $\theta = 0$. Thus in our model, although the international bor-

Series	Canadian data * (1976Q1-2019Q4)		(i) Model with TFP shock		(ii) Model with interest rate shock		(iii) Model with both types of shocks	
	(a)	(b)	(a)	(b)	(a)	(b)	(a)	(b)
Consumption	0.83	0.77	0.17 (0.03)	$0.74 \\ (0.04)$	$0.006 \\ (0.001)$	0.74 (0.04)	0.17 (0.02)	$0.73 \\ (0.04)$
Savings	2.14	0.97	2.71 (0.31)	0.99 (0.00)	0.11 (0.01)	0.99 (0.00)	2.68 (0.27)	$0.99 \\ (0.00)$
Hours worked	1.4	0.86	$0.7 \\ (0.08)$	0.99 (0.00)	$0.03 \\ (0.003)$	0.99 (0.00)	$0.69 \\ (0.07)$	$0.99 \\ (0.00)$
Output	1.42	1	$1.16 \\ (0.13)$	1 (0.00)	$0.047 \\ (0.005)$	$1 \\ (0.00)$	1.1474 (0.11)	$1 \\ (0.00)$
Investment in physical capital	6.63	0.62	5.67 (0.3)	$\begin{array}{c} 0.53 \\ (0.02) \end{array}$	3.26 (0.21)	0.4 (0.04)	6.63 (0.42)	0.47 (0.03)
Trade balance ratio	0.64	-0.12	$1.35 \\ (0.08)$	-0.35 (0.05)	$0.85 \\ (0.05)$	-0.38 (0.04)	1.62 (0.1)	-0.32 (0.04)
Capital productivity	1.84	0.87	$0.94 \\ (0.1)$	0.99 (0.00)	$0.08 \\ (0.009)$	-0.99 (0.00)	$0.93 \\ (0.09)$	$0.99 \\ (0.00)$
Labor productivity	0.75	0.28	$0.48 \\ (0.05)$	$0.99 \\ (0.00)$	0.02 (0.002)	0.99 (0.00)	$0.47 \\ (0.05)$	$0.99 \\ (0.00)$
S - I correlation:	0.59		0.55 (0.02)		0.42 (0.04)		0.49 (0.03)	

Table 1: Statistical Moments: Canadian data and the model economies

Notes: Standard deviations in percent (a) and correlations with output (b).

Small sample standard errors are reported in brackets.

^{*}Output refers to gross domestic product (GDP); consumption to personal expenditures on consumer nondurables, semi-durables, and services; investment in physical capital to business gross fixed investment in nonresidential structures, machinery, and equipment; and hours worked to actual hours at work. Savings equals output minus consumption. Trade balance ratio equals one plus the ratio of net exports to GDP. S - I correlation stands for the contemporaneous correlation between savings and investment in physical capital. All data is from CANSIM by Statistics Canada. Before statistics were reported all series were converted to chained 2012 dollars, seasonally adjusted, deflated by the population, logged, and then detrended via the Hodrick-Prescott (HP) filter.

rowing constraint works to dampen the fluctuations in most of the key macro variables, the high degree of transmission of technology shocks across countries is needed to moderate the variability of investment in physical capital. In Mendoza (1991), on the other hand, the feature of capital adjustment costs works to dampen the variability of most variables, including investment.

Next, we test whether we can lower the volatility of investment in the model enough to better match the data by lowering either σ_g or σ_z^{13} . In Table 3, we report the results when we set $\sigma_g = 0\%$ for three different levels of θ : high, mid, and low. We can see that, while the volatility of the reported variables are dampened by lowering σ_g , for the mid and low values of θ , the variability of investment is still at a counterfactually high level: 24.26% for $\theta = 0.5$ and 44.4% for $\theta = 0$.

Lastly, in Table 4, we present the results from setting $\sigma_z = 0\%$. Overall, the volatility of the reported variables are dampened by lowering σ_z . However, similar to the previous case, for the mid and low values of θ , the variability of investment is still at a counterfactually high level: 41.06% for $\theta = 0.5$ and 82.12% for $\theta = 0$. We can also see that, for a higher values of θ , the volatility of output becomes much smaller compared to the data counterpart.

The main conclusion from the sensitivity analysis in Tables 2, 3, and 4 is as follows. In order for the model to produce a moderate level of volatility of investment in physical capital, we require a high degree of transmission of technology shocks to the world interest rate, and also human capital with borrowing restrictions to mitigate the indirect effects that technology has on investment volatility through the other inputs in the production function. We would not be able to match investment volatility without accounting for both the borrowing restriction and the international transmission of technology shocks.

Our small open economy model is perhaps most similar to Mendoza (1991) in that it has been calibrated to Canada and we consider both technology and interest rate shocks. However, different from us, Mendoza finds that, even for high values for the correlation coefficient between technology and interest rate shocks, he is not able to lower investment volatility enough unless capital adjustment costs are introduced. In order to understand differences between these models, we need to think about what causes fluctuations in physical capital investment, and why they

¹³Note that we cannot set both parameters simultaneously at a very low value because then there will be almost no business cycle fluctuations in the model.

may be different between these two models. These fluctuations are the result of shocks that form temporary gaps (or arbitrage opportunities) between the rate of return to physical capital and the world real interest rate. When a gap forms, investment in physical capital must adjusts to restore equality. For example, a technology shock affects the marginal product of physical capital directly, and also the interest rate via international transmission of technology shocks as discussed earlier. Gaps between rates of return will be larger and investment in physical capital more volatile the smaller the degree of international synchronization of business cycle activity. But this feature applies to both to our setup and Mendoza's setup. However, a technology shock will also affect the rate of return to physical capital indirectly through the other inputs in the production function, since changes in these other inputs affect the marginal product of capital in a complementary way. In our model, these indirect effects will be smaller than in Mendoza's model, leading to additional stability for investment in physical capital. There are two reasons for this. First, compared to Mendoza's model, we have incorporated human capital as an additional input. And by including human capital, the share parameter on labor reduces from 2/3 (= $1 - \alpha$) to 1/6 (= $1 - \alpha - \eta$). This makes the marginal product of physical capital, and therefore physical capital investment, less sensitive to fluctuation in hours worked. Second, the borrowing constraint has the impact of mitigating business cycle fluctuations for most variables, but particularly for human capital investment which is assumed to provide poor collateral for international investors. Although the share parameter on human capital is relatively large, the volatility of the human capital stock is relatively low.

7. Implications of changes in life expectancy on the small open economy

In the next two subsections, we analyze how changes in life expectancy affect, respectively, the steady state levels and business cycle fluctuations of the model variables when the parameters of the model are set at their benchmark values from Section 5. As a numerical exercise, we quantify the contribution of life expectancy to the long-run economic outcome of Canada over the past four decades. As well, we quantify the effect of life expectancy on Canadian business cycle statistics over the same period. It is noting that, while our model is calibrated to Canada, the parameter values are also appropriate for other developed countries, such as those in the OECD. We can also

Table 2: Sensitivity analysis: The degree of transmission of technology shocks, θ , with $\sigma_z = 0.763\%$ and $\sigma_g = 1.567\%$

Series	$\begin{array}{c} \text{Canadian data}^* \\ (1976\text{Q1}2019\text{Q4}) \end{array}$	$\theta = 0.9598$	$\theta = 0.5$	$\theta = 0$
Output	1.42	$1.1474 \\ (0.11)$	$1.509 \\ (0.15)$	2.02 (0.21)
Investment in physical capital	6.63	6.63 (0.42)	47.51 (2.69)	93.01 (5.22)

Notes: Standard deviations in percent. Small sample standard errors are reported in brackets. *Output refers to gross domestic product and investment in physical capital to business gross fixed investment in nonresidential structures, machinery, and equipment. All data is from CANSIM by Statistics Canada. Before statistics were reported all series were converted to chained 2012 dollars, seasonally adjusted, deflated by the population, logged, and then detrended via the Hodrick-Prescott (HP) filter.

Series	$\begin{array}{c} \text{Canadian data}^* \\ (1976\text{Q1}2019\text{Q4}) \end{array}$	$\theta = 0.9598$	$\theta = 0.5$	$\theta = 0$
Output	1.42	$1.1472 \\ (0.11)$	1.4058 (0.14)	1.687 (0.17)
Investment in physical capital	6.63	$5.78 \\ (0.35)$	24.26 (1.47)	44.4 (2.68)

Table 3: Sensitivity analysis: The degree of transmission of technology shocks, θ , with $\sigma_z = 0.763\%$ and $\sigma_g = 0\%$

Notes: Standard deviations in percent. Small sample standard errors are reported in brackets. *Output refers to gross domestic product and investment in physical capital to business gross fixed investment in nonresidential structures, machinery, and equipment. All data is from CANSIM by Statistics Canada. Before statistics were reported all series were converted to chained 2012 dollars, seasonally adjusted, deflated by the population, logged, and then detrended via the Hodrick-Prescott (HP) filter.

see in Figure 4 that changes in the life expectancy over the time period being investigated are similar between Canada and other OECD countries.

7.1. The effect of life expectancy on the steady state

As panel A of Figure 5 shows, increased life expectancy is found have a positive and notable impact on the steady state level of output per effective labor. For example, in Canada, between 1976 and 2018, life expectancy at birth increased from around 74 years to 82 years, which translates to an estimated 5.5% increase in the long-run output per effective labor.

Our model suggests that long-run output per effective labor is more sensitive to changes in life expectancy at low levels of life expectancy than it is at high levels. This can be seen more clearly

Series	$\begin{array}{c} {\rm Canadian\ data^*}\\ (1976{\rm Q1-}2019{\rm Q4}) \end{array}$	$\theta = 0.9598$	$\theta = 0.5$	$\theta = 0$
Output	1.42	$0.046 \\ (0.004)$	$\begin{array}{c} 0.573 \ (0.05) \end{array}$	$1.1456 \\ (0.11)$
Investment physical capital	6.63	$3.301 \\ (0.17)$	41.06 (2.2)	82.12 (4.39)

Table 4: Sensitivity analysis: The degree of transmission of technology shocks, θ , with $\sigma_z = 0\%$ and $\sigma_g = 1.567\%$

Notes: Standard deviations in percent. Small sample standard errors are reported in brackets. *Output refers to gross domestic product and investment in physical capital to business gross fixed investment in nonresidential structures, machinery, and equipment. All data is from CANSIM by Statistics Canada. Before statistics were reported all series were converted to chained 2012 dollars, seasonally adjusted, deflated by the population, logged, and then detrended via the Hodrick-Prescott (HP) filter.



Figure 4: Average life expectancy at birth. Life expectancy at birth is defined as the average number of years a newborn is expected to live if mortality patterns at the time of its birth remain constant throughout its life. Data are from the World Development Indicators by the World Bank.

from panel B of Figure 5, which illustrates the percentage changes in the steady state output per effective labor as life expectancy increases. We can see that increased life expectancy has positive but diminishing marginal effect on long-run output per effective labor. This result follows from the model's feature of diminishing returns to human capital.

Finally, panels C through F of Figure 5 show how the steady state levels of consumption, aggre-

gate hours worked, human capital, and physical capital change following changes in life expectancy. We note that, in our model, consistent with standard theory of human capital, increased life expectancy makes investment in human capital more profitable and thus encourages its accumulation. Given the complementarity between the two types of capital in the production function, the increase in the stock of human capital then leads to an increase in the stock of physical capital. As can be seen from panel D, average hours worked per period decreases, albeit rather moderately, as life expectancy increases. This suggests that increased life expectancy allows the agents to enjoy more leisure per period. The rise in both leisure and consumption are due to having greater lifetime wealth. Given the foregoing changes in human capital, physical capital, and hours worked, the output level of the domestic economy rises as life expectancy increases.

7.2. The effect of life expectancy on business cycle fluctuations

Figure 6 below shows the effect of changes in life expectancy on the standard deviations of the model variables when the parameters are set at their benchmark values from Section 5. We can see that volatility of consumption, human capital, and physical capital all decrease as life expectancy increases. In the model, an increase in life expectancy raises the effective time preference of the domestic economy and the agents become more patient. As a result, the volatility of human capital decreases as life expectancy increases, shown in panels C. Given the complementarity between the two types of capital, the volatility of physical capital also decreases as life expectancy increases as shown in panel D. Shown in panel F, the volatility of trade balance ratio is found to increase as life expectancy increases.

Next, as shown in panel E, the volatility of average hours worked per period increases as life expectancy increases. When life expectancy is low, the agents have no luxury to increase or adjust their leisure hours and hence there is less volatility in the hours worked. On the other hand, when life expectancy is high, the agents become more flexible to increase their leisure hours and thus there is more volatility in the hours worked. As can be seen from panel B, following the decreased volatilities of physical and human capital and the increased volatility of hours worked, output ends up becoming more volatile as life expectancy increases. Panel A shows that consumption becomes less volatile as life expectancy increases, suggesting an improvement in consumption smoothing.



Figure 5: The steady state and life expectancy



Figure 6: Life expectancy and standard deviation of the model variables

Followed from the model's feature of diminishing returns to human capital, we find that changes in life expectancy will have a greater effect—in terms of magnitude—on the volatility of the above mentioned aggregate variables the lower is the initial level of life expectancy. This can be seen Figure 7, which illustrates the percentage changes in the standard deviations of consumption and average hours worked per period due to changes in life expectancy. Although we report the results for the given two variables only, we get the similar pattern for the other variables as well: As life expectancy increases, the marginal effects it has on the standard deviations of the model variables decrease in magnitude.

Lastly, our numerical exercise reveals that, in Canada, due to the change in life expectancy between 1976 and 2018, the volatility of consumption has decreased by an estimated 1.55%. On the other hand, the volatility of hours worked, output, and trade balance ratio have increased by an estimated 0.3%, 0.043%, and 0.277%, respectively.



Figure 7: Life expectancy and marginal change in standard deviation of the model variables

8. Conclusion

We develop a standard small open economy RBC model driven by neutral productivity shocks and interest-rate shocks that is consistent with the main empirical facts of economic fluctuations in an actual open economy. The model is calibrated to the Canadian economy. As in the data, the model is consistent with the low volatility of consumption and large volatility of investment relative to that of output. The model is also able to capture the high positive correlations of consumption, savings, hours worked, and investment in physical capital with output; the sufficiently countercyclical trade balance ratio; and the procyclical interest rate. The observed positive correlation between domestic savings and investment in physical capital is also well captured by the model. Further, the impulse responses of aggregate variables to a TFP shock as well as an interest-rate shock are found to be consistent with the standard results in the literature. Given its finite lifetimes feature, our framework allows us to study the effects of life expectancy on the cyclical fluctuations of the aggregate variables within the context of a small open economy. Our results reveal that the fluctuations in physical capital, human capital, and consumption all decrease as life expectancy increases. On the other hand, the fluctuation of hours worked, trade balance ratio, and output are found to increase with life expectancy. We also find that changes in life expectancy will have a greater effect—in terms of magnitude—on the volatility of the above mentioned variables the lower is the initial level of life expectancy. Using our model, we have quantified how changes in life expectancy have impacted the long-run economic outcome and business cycle fluctuations of Canada over the past forty years.

References

- [1] Alessandrini, D., Kosempel, S., and Stengos, T. (2015). The business cycle human capital accumulation nexus and its effect on hours worked volatility. *Journal of Economic Dynamics and Control*, 51(C):356–377.
- [2] Barro, R. J., Mankiw, N. G., and Sala-i Martin, X. (1995). Capital Mobility in Neoclassical Models of Growth. American Economic Review, 85(1):103–115.
- [3] Baxter, M. and Kouparitsas, M. A. (2005). Determinants of business cycle comovement: a robust analysis. Journal of Monetary Economics, 52(1):113–157.
- [4] Blanchard, O. (1985). Debt, deficits, and finite horizons. Journal of Political Economy, 93(2):223-47.
- [5] Burstein, A., Kurz, C., and Tesar, L. (2008). Trade, production sharing, and the international transmission of business cycles. *Journal of Monetary Economics*, 55(4):775–795.
- [6] Cardia, E. (1991). The dynamics of a small open economy in response to monetary, fiscal, and productivity shocks. *Journal of Monetary Economics*, 28(3):411–434.
- [7] Clark, T. and van Wincoop, E. (2001). Borders and business cycles. Journal of International Economics, 55(1):59–85.
- [8] Cooley, T. and Hansen, G. (1989). The inflation tax in a real business cycle model. American Economic Review, 79(4):733–48.
- [9] Correia, I., Neves, J. C., and Rebelo, S. (1995). Business cycles in a small open economy. *European Economic Review*, 39(6):1089–1113.
- [10] Dantas Guimarães, S. and Ferreira Tiryaki, G. (2020). The impact of population aging on business cycles volatility: International evidence. *The Journal of the Economics of Ageing*, 17:100285.
- [11] di Giovanni, J. and Levchenko, A. A. (2010). Putting the Parts Together: Trade, Vertical Linkages, and Business Cycle Comovement. American Economic Journal: Macroeconomics, 2(2):95–124.

- [12] Frankel, J. A. and Rose, A. K. (1998). The endogenity of the optimum currency area criteria. The Economic Journal, 108(449):1009–1025.
- [13] Frenkel, J. A. and Razin, A. (1986). Fiscal policies in the world economy. *Journal of Political Economy*, 94(3):564–94.
- [14] Giovanni, J. d. and Levchenko, A. A. (2009). Trade Openness and Volatility. The Review of Economics and Statistics, 91(3):558–585.
- [15] Gomme, P. (1993). Money and growth revisited: Measuring the costs of inflation in an endogenous growth model. *Journal of Monetary Economics*, 32(1):51–77.
- [16] Gundlach, E. (1995). The role of human capital in economic growth: New results and alternative interpretations. Weltwirtschaftliches Archiv, 131(2):383–402.
- [17] Hansen, G. D. (1985). Indivisible labor and the business cycle. Journal of Monetary Economics, 16(3):309–327.
- [18] Jaimovich, N. and Siu, H. E. (2009). The young, the old, and the restless: Demographics and business cycle volatility. *American Economic Review*, 99(3):804–26.
- [19] Kydland, F. E. and Prescott, E. C. (1982). Time to Build and Aggregate Fluctuations. *Econometrica*, 50(6):1345– 1370.
- [20] Leiderman, L. and Razin, A. (1989). Current Account Dynamics: The Role of Real Shocks. IMF Working Papers 1989/080, International Monetary Fund.
- [21] Letendre, M.-A. (2004). Capital utilization and habit formation in a small open economy model. Canadian Journal of Economics, 37(3):721–741.
- [22] Letendre, M.-A. and Luo, D. (2007). Investment-specific shocks and external balances in a small open economy model. *Canadian Journal of Economics*, 40(2):650–678.
- [23] Lugauer, S. (2012a). Demographic change and the great moderation in an overlapping generations model with matching frictions. *Macroeconomic Dynamics*, 16(5):706–731.
- [24] Lugauer, S. (2012b). Estimating the Effect of the Age Distribution on Cyclical Output Volatility Across the United States. The Review of Economics and Statistics, 94(4):896–902.
- [25] Mankiw, N. G., Romer, D., and Weil, D. N. (1992). A Contribution to the Empirics of Economic Growth. The Quarterly Journal of Economics, 107(2):407–437.
- [26] McGrattan, E. and Prescott, E. (2001). Is the stock market overvalued? NBER Working Papers 8077, National Bureau of Economic Research, Inc.
- [27] Mendoza, E. and Uribe, M. (2000). Devaluation risk and the business-cycle implications of exchange-rate management. Carnegie-Rochester Conference Series on Public Policy, 53(1):239–296.
- [28] Mendoza, E. G. (1991). Real Business Cycles in a Small Open Economy. American Economic Review, 81(4):797–818.
- [29] Nason, J. M. and Rogers, J. H. (2006). The present-value model of the current account has been rejected: Round up the usual suspects. *Journal of International Economics*, 68(1):159–187.

- [30] Neumeyer, P. A. and Perri, F. (2005). Business cycles in emerging economies: the role of interest rates. *Journal of Monetary Economics*, 52(2):345–380.
- [31] Obstfeld, M. (1981). Macroeconomic policy, exchange-rate dynamics, and optimal asset accumulation. Journal of Political Economy, 89(6):1142–1161.
- [32] Obstfeld, M. (1990). Intertemporal dependence, impatience, and dynamics. Journal of Monetary Economics, 26(1):45–75.
- [33] Prescott, E. C. (1986). Theory ahead of business-cycle measurement. Carnegie-Rochester Conference Series on Public Policy, 25:11–44.
- [34] Psacharopoulos, G. and Patrinos, H. A. (2018). Returns to investment in education: a decennial review of the global literature. *Education Economics*, 26(5):445–458.
- [35] Schmitt-Grohe, S. (1998). The international transmission of economic fluctuations:: Effects of U.S. business cycles on the Canadian economy. *Journal of International Economics*, 44(2):257–287.
- [36] Schmitt-Grohe, S. and Uribe, M. (2001). Stabilization policy and the costs of dollarization. Journal of Money, Credit and Banking, 33(2):482–509.
- [37] Schmitt-Grohe, S. and Uribe, M. (2003). Closing small open economy models. Journal of International Economics, 61(1):163–185.
- [38] Senhadji, A. S. (2003). External shocks and debt accumulation in a small open economy. *Review of Economic Dynamics*, 6(1):207–239.
- [39] Tserenkhuu, T. and Kosempel, S. (2023). Open economy neoclassical growth models and the role of life expectancy. The B.E. Journal of Macroeconomics, 23(2):1057–1092.
- [40] Uhlig, H. (2001). A Toolkit for Analysing Nonlinear Dynamic Stochastic Models Easily. In Computational Methods for the Study of Dynamic Economies. Oxford University Press.
- [41] Uribe, M. (1997). Exchange-rate-based inflation stabilization: The initial real effects of credible plans. Journal of Monetary Economics, 39(2):197–221.
- [42] Uzawa, H. (1968). Time preference, the consumption function, and optimum asset holdings. Value, Capital, and Growth: Papers in Honour of Sir John Hicks, pages 485–504.
- [43] Yaari, M. E. (1965). Uncertain lifetime, life insurance, and the theory of the consumer. The Review of Economic Studies, 32(2):137–150.

Appendix: Statistics Canada (CANSIM) Data Series Label Retrieval Codes

Series title	Series label
1. Gross domestic product at market prices	V62305752
2. Household consumption of semi-durable goods	V62305727
3. Household consumption of non-durable goods	V62305728
4. Household consumption of services	V62305729
5. Business investment in non-residential structures, machinery and equipment	V62305735
6. Exports of goods and services	V62305745
7. Imports of goods and services	V62305748
8. Total actual hours worked, all industries	V4391505
9. Population	V1
10. Total expenditures on education $(1976-1999)$	V1996800
11. Public and private elementary and secondary education expenditures (2000–2019)	V1200358
12. Expenditures of universities and degree granting colleges (2000–2019)	V80711164
13. GDP Implicit price index $(2012=100)$	V62471023

Notes: Most data is quarterly (1976Q1–2019Q4), in chained 2012 dollars, and seasonally adjusted at annual rates. The exception is data used to construct a series for human capital investment, which is availably only annually and in current dollars. These values were converted to constant 2012 dollars using the GDP deflator and then to quarterly frequency by interpolating linearly between points.